ORIGINAL RESEARCH

# Diagnosis of Tuberculous Pericarditis in Zhejiang, China: A Diagnostic Prediction Model Based on LASSO Logistic Regression

Xiaoqun Xu<sup>1,</sup>\*, Xiao Liu<sup>2,</sup>\*, Chao Yang<sup>2,</sup>\*, Long Cai<sup>1</sup>, Libin Liu<sup>1</sup>, Tielong Chen<sup>3</sup>, Houyong Zhu<sup>1</sup>, Hui Wei<sup>1</sup>

<sup>1</sup>Centre of Laboratory Medicine, Hangzhou Red Cross Hospital, Hangzhou, Zhejiang, People's Republic of China; <sup>2</sup>The Fourth School of Clinical Medicine, Zhejiang Chinese Medical University, Hangzhou, Zhejiang, People's Republic of China; <sup>3</sup>Department of Cardiology, Hangzhou TCM Hospital Affiliated to Zhejiang Chinese Medical University, Hangzhou, Zhejiang, People's Republic of China

\*These authors contributed equally to this work

Correspondence: Hui Wei, Hangzhou Red Cross Hospital, No. 208 East Huancheng Road, Hangzhou, 310003, People's Republic of China, Email weihui-hzred@qq.com; Houyong Zhu, Department of Cardiology, Hangzhou TCM Hospital Affiliated to Zhejiang Chinese Medical University, No. 453 Stadium Road, Hangzhou, 310007, People's Republic of China, Email houyongzhu@foxmail.com

**Background and Aims:** Tuberculous pericarditis (TBP) is a severe, life-threatening complication, yet its diagnosis is highly challenging due to the lack of sufficient diagnostic tools. The aim of this study was to develop and validate a diagnostic prediction model suitable for primary healthcare institutions to predict the risk of TBP.

**Methods:** We collected detailed medical histories, imaging examination results, laboratory test data, and clinical characteristics of patients and used the Least Absolute Shrinkage and Selection Operator (LASSO) technique combined with logistic regression analysis to construct a predictive model. The diagnostic efficacy of the model was assessed using the Receiver Operating Characteristic (ROC) curve, calibration curve, and Decision Curve Analysis (DCA).

**Results:** A total of 304 patients were included in the study, with a median age of 64 years, of which 144 were diagnosed with tuberculous pericarditis. Patients were randomly assigned to the training and validation sets in a 7:3 ratio. LASSO logistic regression analysis revealed that weight loss (P=0.011), body mass index (BMI) (P=0.061), history of tuberculosis (P=0.022), history of dust exposure (P=0.03), moderate to severe kidney disease (P=0.005), erythrocyte sedimentation rate (ESR) (P=0.084), and B-type natriuretic peptide (BNP) (P<0.001) are independent risk factors for TBP. Based on these factors, we constructed a nomogram with an Area Under the Receiver Operating Characteristic Curve (AUC) of 0.757 in both the training and validation sets, indicating high discriminative ability of the model. Calibration curve analysis showed good consistency of the model. DCA results indicated that the model has significant clinical application value when the threshold probability is set between 1–100% (training set) and 30–100% (validation set).

**Conclusion:** We successfully developed a nomogram model for predicting tuberculous pericarditis, which can assist clinicians in improving diagnostic accuracy and reducing misdiagnoses and missed diagnoses in primary healthcare settings.

**Plain Language Summary:** Imagine you have a tool that helps doctors figure out if someone has a serious heart issue called tuberculous pericarditis, which is tough to detect. Our team collected data from 304 patients, looking at everything from their medical history to lab results. We used a smart method to pinpoint key risk factors like weight loss, body mass index, past tuberculosis, exposure to dust, moderate or severe kidney disease, a measure of inflammation called erythrocyte sedimentation rate, and a heart failure marker known as B-type natriuretic peptide. From there, we crafted a simple scoring tool that predicts the likelihood of having this heart problem. When we put our tool to the test, it did a great job, especially when we set the risk level just right. This means we have developed a helpful guide for doctors, especially in places with limited resources, to diagnose this condition more accurately and avoid mistakes. In simple terms, our research has led to a better way to spot a dangerous heart condition that can be tricky to find. This not only helps patients get the right treatment but also raises awareness about the importance of medical research and its impact on public health.

Keywords: tuberculous pericarditis, diagnostic prediction model, primary healthcare, LASSO logistic regression

© 2025 Xu et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/terms.ph you hereby accept the firms. Non-commercial uses of the work are permitted without any further permission from Dove Medical Press Limited, provided the work is properly attributed. For permission for commercial use of this work, please see paragraphs 4.2 and 5 of our Terms (http://www.dovepress.com/terms.ph).

4681

#### **Graphical Abstract**



## Introduction

Tuberculosis is a very serious disease, often becoming the leading cause of death from a single infectious disease worldwide. Over the past three years, it was dethroned by COVID-19, but now it is poised to reclaim its position at the top.<sup>1</sup> Among the various forms of tuberculosis, tuberculous pericarditis (TBP) stands out as a particularly severe manifestation. Without prompt and effective treatment, TBP can lead to severe complications such as constrictive pericarditis, cardiac tamponade, and even death.<sup>2,3</sup> Due to its rapid progression and potential for fatality, in some developing countries, the mortality rate among patients with untreated TBP within six months of onset ranges from 17% to 40%.<sup>4</sup> In Zhejiang Province, China, a recent survey report<sup>5</sup> indicated that the region's implementation of tuberculosis health management service projects was generally good, partly due to the application of new diagnostic technologies such as the molecular diagnostic technique Xpert MTB/RIF. However, despite these advancements, the rapid and accurate diagnosis of TBP still faces significant challenges. The low number of mycobacteria in pericardial effusion not only poses a problem for traditional microbiological methods, such as acid-fast bacilli (AFB) smears and culture methods, which suffer from low sensitivity,<sup>6</sup> but also limits the effectiveness of molecular diagnostic techniques like Xpert MTB/RIF. These technologies require sufficient pathogen DNA to produce reliable results, and the high cost of repeat testing can exacerbate the economic burden on patients. Pericardial biopsy, while a diagnostic option, is an invasive procedure that demands specialized clinical skills and may impose both psychological and physical burdens on patients.

In primary healthcare facilities in Zhejiang Province, the diagnosis of TBP faces significant challenges due to the uneven distribution of medical resources. Particularly in small primary healthcare service sites, the lack of key imaging tools such as radiological equipment, along with a deficiency in professional knowledge about TBP diagnosis among medical staff, makes rapid and accurate diagnosis even more difficult. This situation not only limits the effective management of tuberculosis in these areas but is also commonly found in primary healthcare facilities in other countries.<sup>7</sup> There is an urgent need for a clinical diagnostic model based on easily accessible data such as demographic characteristics, clinical presentations, and common laboratory results to assist clinicians in rapidly and effectively diagnosing TBP.

Addressing this need, we conducted a retrospective study and developed a diagnostic prediction model presented in the form of a nomogram. In this study, we integrated clinical characteristics, laboratory indicators, and imaging data of patients to develop a novel diagnostic prediction model. The model is designed to provide an effective diagnostic tool for clinicians, especially in primary healthcare institutions and healthcare environments with limited medical resources, to improve the diagnostic accuracy of tuberculous pericarditis (see Graphical Abstract for a visual representation of the model's framework).

## Methods

#### Ethical Statement and Data Source

This study, a single-center retrospective cohort study conducted at Hangzhou Red Cross Hospital from January 2015 to August 2023, aimed to establish a predictive model for the diagnosis of tuberculous pericarditis. Hangzhou Red Cross Hospital serves as a provincial tuberculosis treatment center, treating patients with tuberculosis from various regions of Zhejiang Province, China. The study involved patients with tuberculous pericarditis and non-specific pericarditis who were consecutively hospitalized for treatment during this period. In accordance with the ethical standards of the Declaration of Helsinki, the World Medical Association, and the Council for International Organizations of Medical Sciences (CIOMS) guidelines for international ethics in human biomedical research, the Ethics Committee of Hangzhou Red Cross Hospital reviewed and approved this study. A waiver for the requirement of informed consent was granted by the Ethics Committee due to the retrospective nature of the study (Ethics Application Number: 2024YS90). The study strictly adheres to the principles outlined in the Declaration of Helsinki and ensures the protection of participants' rights and welfare. We pledge to keep confidential all patient information gathered from the electronic medical records database. The raw data supporting the results of this study have been uploaded to Zenodo and are accessible upon reasonable request at the following link: <a href="https://zenodo.org/records/13822455">https://zenodo.org/records/13822455</a>.

#### **Patient Selection**

Inclusion Criteria: The diagnosis of TBP is based on any of the following criteria: 1. Mycobacterium tuberculosis is isolated from pericardial effusion or pericardial biopsy specimens, confirmed by positive acid-fast staining or tuberculosis culture. 2. Histological examination of pericardial biopsy samples shows granulomatous inflammation. 3. In the presence of clinical and radiological evidence of tuberculosis, there is a positive response to standardized anti-tuberculosis treatment, and Mycobacterium tuberculosis is isolated from sputum or non-pericardial exudate.<sup>8</sup> The diagnosis of non-specific pericarditis is based on: patients presenting with clinical manifestations of pericarditis, but without clear evidence of pathogen infection in clinical tests, and postoperative pathological diagnosis shows non-specific pericardial inflammatory changes. Exclusion Criteria: Patients with incomplete clinical data.

## Data Collection

During the model development process, we documented the following age-adjusted Charlson Comorbidity Index (CCI)<sup>9</sup> and a series of clinically common indicators that may suggest TBP: demographic characteristics: age, gender, height, and weight (used to calculate body mass index (BMI)), marital status, level of education, immigration status; lifestyle factors: smoking status, history of alcohol consumption; clinical symptoms: chest pain, fever, night sweats, cough, difficulty breathing, weight loss; past medical history: chronic infections, history of tuberculosis, history of dust exposure, hypertension, diabetes, connective tissue diseases, liver diseases, moderate to severe kidney diseases, tumors, leukemia, lymphoma, acquired immune deficiency syndrome (AIDS); clinical examination results: pericardial friction rub, electrocardiogram changes, amount of pericardial effusion; laboratory test results: neutrophil count, lymphocyte count, monocyte count, hemoglobin level, platelet count, erythrocyte sedimentation rate (ESR), C-reactive protein (CRP), creatine kinase (CK), creatine kinase isoenzyme MB (CK-MB), B-type natriuretic peptide (BNP). All biochemical indicators were selected and recorded based on the patients' first set of biochemical test results after admission. Detailed diagnostic criteria can be found in <u>Appendix 1</u>.

## Model Development and Validation

Initially, the dataset was randomly divided into a training set and a validation set in a 7:3 ratio. The training set was used to develop the diagnostic model and prepare the nomogram, while the validation set was utilized for the model's verification process. Subsequently, the Least Absolute Shrinkage and Selection Operator (LASSO) method was employed for preliminary variable screening. To compare the fit of seven logistic regression modeling strategies, we evaluated them based on the Akaike Information Criterion (AIC) and the Bayesian Information Criterion (BIC). The results revealed that, aside from the enter method, the remaining six strategies showed consistency in model fit, and all displayed lower AIC and BIC values compared to the enter method (Supplementary Table 1). Consequently, we selected any one of these six strategies for modeling in this study. Ultimately, we employed the backward likelihood ratio (LR) method for multivariate logistic regression analysis to identify independent risk factors in the model. Based on the results of the multivariate regression model, a nomogram for diagnosing tuberculous pericarditis was constructed. The model's discriminative ability was assessed using the Area Under the Receiver Operating Characteristic Curve (AUC). The Hosmer-Lemeshow test was used to evaluate the model's calibration.

## Clinical Decision Analysis and Model Application

To evaluate the clinical utility of the model, we used clinical decision curve analysis (DCA) to assess the model's performance in both the training and validation sets. Based on the results of the validation set and the Youden index, the total score of the nomogram was used to stratify patients with tuberculous pericarditis into low-risk, medium-risk, and high-risk categories. In clinical practice, due to the lack of precise diagnostic tools, physicians often rely on clinical experience to diagnose tuberculous pericarditis. To assess the practical application value of the model, we compared the model's risk assessment with that of physicians based on clinical experience, including: misdiagnosis - analyzing non-TBP patients in the validation set to evaluate the proportion of TBP misdiagnosed by physicians' clinical experience and the proportion misdiagnosed as TBP based on the model's high-risk group threshold, and comparing the misdiagnosis rates between the two; missed diagnosis analyzing TBP patients in the validation set to evaluate the proportion of TBP missed by physicians' clinical experience and the proportion incorrectly excluded as TBP based on the model's low-risk group threshold, and comparing the missed diagnosis rates between the two; overuse of anti-tuberculosis treatment - assessing non-TBP patients in the validation set to observe the proportion given anti-tuberculosis treatment in actual clinical practice.

## Statistical Analysis

In this study, continuous variables were summarized using medians and interquartile ranges (IQR), and the Mann–Whitney *U*-test was utilized to compare statistical differences between the two groups. For categorical variables, we conducted analyses using Pearson's chi-squared test. To address missing data, we employed the Expectation-Maximization (EM) algorithm for estimating and imputing missing values, thereby enhancing the completeness of the data analysis. Odds ratios (OR) and their corresponding 95% confidence intervals (CI) were obtained from multivariate logistic regression models. Data analysis was primarily conducted using SPSS software version 27 and STATA software version 15.

## Results

## **Patient Characteristics**

A total of 304 patients were ultimately included in this study, with a median age of 64 years, ranging from a minimum age of 16 years, and 29.28% were female (Table 1). They were randomly assigned to the training and validation sets in a 7:3 ratio (Supplementary Figure 1). Comparisons between the training and validation sets revealed no statistically significant differences in most indicators, except for a history of dust exposure, moderate to severe kidney disease, and weight loss, which showed significant differences between the two groups. Compared to patients with non-specific pericarditis, those with TBP in both the training and validation sets exhibited the following characteristics: lower BMI, a higher proportion of weight loss, lower hemoglobin levels, and a higher proportion of tuberculosis history and dust exposure history (Supplementary Table 2). For the handling of missing data, we used the EM algorithm to estimate and impute the following variables: BMI, neutrophils, lymphocytes, monocytes, hemoglobin, platelets, ESR, CRP, BNP, CK, CK-MB, electrocardiogram, and pericardial effusion.

Table I Baseline Features of Training and Validation Datas
--

Characteristics	Whole Cohort (N=304)	Training Dataset (N=222)	Validation Dataset (N=82)	P Value
Age, median (quartile), years	64 (54–71)	64 (54–71)	65 (54–72)	0.204
Female, no. (%)	89 (29.28)	65 (29.3)	24 (29.3)	0.999
BMI, median (quartile), kg/m <sup>2</sup>	17.65 (15.77–19.44)	17.68 (15.81–19.47)	17.49 (15.43–19.20)	0.467
Marital status, no. (%)				0.204
Married	275 (90.5)	203 (91.4)	72 (87.8)	
Unmarried	10 (3.3)	6 (2.7)	4 (4.9)	
Divorce	5 (1.6)	5 (2.3)	0 (0)	
Widowed	14 (4.6)	8 (3.6)	6 (7.3)	
Education status, no. (%)				0.908
≥ High school	47 (15.46)	34 (15.3)	13 (15.9)	
< High school	257 (84.54)	188 (84.7)	69 (84.1)	
Migrant, no. (%)	3 (0.99)	3 (1.4)	0 (0)	0.290
Smoking, no. (%)	92 (30.26)	69 (31.1)	23 (28)	0.610
Consumed alcohol no. (%)	76 (25)	60 (27)	16 (19.5)	0.179
Chest pain, no. (%)	43 (14.14)	28 (12.6)	15 (18.3)	0.207
Fever, no. (%)	108 (35.5)	75 (33.8)	33 (40.2)	0.296
Night sweat, no. (%)	30 (9.87)	22 (9.9)	8 (9.8)	0.968
Cough, no. (%)	157 (51.64)	121 (54.5)	36 (43.9)	0.101
Dyspnea, no. (%)	248 (81.58)	181 (81.5)	67 (81.7)	0.972
Weight loss, no. (%)	58 (19.08)	36 (16.2)	22 (26.8)	0.037
Chronic infection, no. (%)	37 (12.17)	26 (11.7)	( 3.4)	0.687
Pericardial friction sound, no. (%)	I (0.33)	I (0.5)	0 (0)	0.543
Changes in electrocardiogram, no. (%)	4 (1.3)	2 (0.9)	2 (2.4)	0.403
Pericardial effusion, no. (%)				0.078
< 5mm	172 (56.58)	129 (58.1)	43 (52.4)	
5–10mm	80 (26.32)	50 (22.5)	30 (36.6)	
10–20mm	38 (12.5)	30 (13.5)	8 (9.8)	
>20mm	11 (3.62)	10 (4.5)	I (1.2)	
Neutrophils, median (quartile), ×10 <sup>9</sup> /L	3.60 (2.70-4.76)	3.60 (2.78–4.70)	3.54 (2.38-4.90)	0.261
Lymphocyte, median (quartile), ×10 <sup>9</sup> /L	0.80 (0.59–1.10)	0.80 (0.60–1.10)	0.8 (0.0.54–1.06)	0.642
Monocyte, median (quartile), ×10 <sup>9</sup> /L	0.50 (0.40–0.60)	0.50 (0.40–0.64)	0.50 (0.31–0.60)	0.277
Hemoglobin, median (quartile), g/L	119.50 (109–131)	120 (110–131)	119 (108–130.50)	0.535
Platelet, median (quartile), ×10 <sup>9</sup> /L	176 (130.75–246.50)	178.5 (132.25–243.21)	165.5 (129.75–269.75)	0.665

(Continued)

Table I	(Continued).
---------	--------------

Characteristics	Whole Cohort (N=304)	Training Dataset (N=222)	Validation Dataset (N=82)	P Value
ESR, median (quartile), mm/h	38.76 (23.11–57.89)	38 (23.34–58)	41.10 (22–57.50)	0.916
CRP, median (quartile), mg/L	20.6 (9.91–46.44)	21.76 (9.48-48.18)	19.57 (12.16–34.18)	0.474
CK, median (quartile), U/L	55 (35–75.58)	55.66 (34–79)	51.5 (35.75–69.02)	0.730
CK-MB, median (quartile), U/L	10 (8–13)	10.25 (8–13)	10 (7–11.87)	0.190
BNP, median (quartile), pg/mL	250.5 (145.19–361.25)	250 0.5 (148.75–357.81)	249 (129–401.75)	0.797
Tuberculosis history, no. (%)	35 (11.51)	24 (10.8)	(13.4)	0.528
Dust exposure history, no. (%)	31 (10.20)	15 (6.8)	16 (19.5)	0.001
Hypertension, no. (%)	74 (24.34)	50 (22.5)	24 (29.3)	0.224
Diabetes, no. (%)	46 (15.13)	36 (16.2)	10 (12.2)	0.385
Connective tissue disease, no. (%)	3 (0.99)	I (0.5)	2 (2.4)	0.120
Liver dysfunction, no. (%)	82 (26.97)	66 (29.7)	16 (19.5)	0.075
Moderate or severe nephropathy, no. (%)	49 (16.12)	29 (13.1)	20 (24.4)	0.017
Cancer, no. (%)	9 (2.96)	8 (3.6)	(1.2)	0.276
Leukemia, no. (%)	0 (0)	0 (0)	0 (0)	-
Lymphoma, no. (%)	0 (0)	0 (0)	0 (0)	-
AIDS, no. (%)	0 (0)	0 (0)	0 (0)	-

Note: Values are numbers (%) or medians (quartile).

Abbreviations: AIDS, acquired immune deficiency syndrome; BMI, body mass index; BNP, B-type natriuretic peptide; CK, creatine kinase; CK-MB, creatine kinase isoenzyme MB; CRP, C-reactive protein; ESR, erythrocyte sedimentation rate; TBP, Tuberculous pericarditis.

The proportions of missing data imputed for these variables were 21.7%, 0.3%, 0.3%, 0.3%, 0.3%, 16.4%, 0.7%, 15.5%, 15.5%, 15.5%, 0.7%, and 1.0%, respectively (Supplementary Table 3).

## Variable Selection and Nomogram Construction

Based on the principle of the maximum allowable lambda value within one standard error of the minimum loss, after triple cross-validation of the LASSO coefficient distribution for 39 features, lambda=13.145167 was selected as the standard for this LASSO regression. A total of 8 variables were retained, including weight loss, BMI, history of tuberculosis, history of dust exposure, moderate to severe kidney disease, platelets, ESR, and BNP (Supplementary Figure 2). The aforementioned variables were further subjected to multivariate logistic regression, ultimately determining that weight loss [OR, 3.131, (95% CI, 1.305–7.511), P=0.011], BMI [OR, 0.886, [95% CI, 0.780–1.006), P=0.061], previous tuberculosis history [OR, 3.529, (95% CI, 1.197–10.406), P=0.022], history of dust exposure [OR, 4.036, (95% CI, 1.144–14.245), P=0.030], moderate to severe kidney disease [OR, 4.121, (95% CI, 1.539–11.031), P=0.005], ESR [OR, 1.011, (95% CI, 0.999–1.023), P=0.084], and BNP [OR, 0.997, (95% CI, 0.995–0.999), P<0.001] were independently associated factors for diagnosing tuberculous pericarditis (Table 2). These factors were used to construct a nomogram model to visualize the model (Figure 1).

#### Model Validation

The AUC of the diagnostic model for TBP was 0.757 (95% CI, 0.694-0.820) in the training set and 0.757 (95% CI, 0.651-0.862) in the validation set (Figure 2A and B). The Hosmer–Lemeshow goodness-of-fit test showed P=0.863 in the training dataset and P=0.084 in the validation dataset, and the calibration curves demonstrated good consistency between predicted and observed values (Figure 2C and D).

Predictors	OR (95% CI)	P Value	
BMI	0.886 (0.780-1.006)	0.061	
Weight loss	3.131 (1.305–7.511)	0.011	
Tuberculosis history	3.529 (1.197–10.406)	0.022	
Dust exposure history	4.036 (1.144–14.245)	0.030	
Moderate or severe nephropathy	4.121 (1.539–11.031)	0.005	
ESR	1.011 (0.999–1.023)	0.084	
BNP	0.997 (0.995–0.999)	<0.001	
Platelet	-	0.721	

 Table 2 Multivariate Logistic Regression

Abbreviations: BMI, body mass index; BNP, B-type natriuretic peptide; ESR, erythrocyte sedimentation rate; CI, confidence intervals; OR, odds ratio.

## **Clinical Application**

Results from the DCA indicated that the model achieved the maximum clinical net benefit when the threshold probabilities were set from 1–100% in the training cohort and 30–100% in the validation cohort (Figure 3A and B). Based on the DCA from the validation set, we first identified the optimal threshold for low risk at 11 points and then set the optimal threshold between medium and high risk at 14 points according to the Youden index, thereby stratifying



Figure I Nomogram Developed from the LASSO Logistic Regression Model.

Abbreviations: BMI, body mass index; BNP, B-type Natriuretic Peptide; ESR, erythrocyte sedimentation rate; TBP, Tuberculous pericarditis.



Figure 2 Analysis of the ROC and Calibration Plots. (A) Model performance in diagnosing tuberculous pericarditis within the training set. (B) Model performance in diagnosing tuberculous pericarditis within the validation set. (C) Calibration of the model for diagnosing tuberculous pericarditis within the training set. (D) Calibration of the model for diagnosing tuberculous pericarditis within the validation set. The sawtooth pattern in red above represents patients diagnosed with tuberculous pericarditis (coded as I), while the sawtooth pattern below represents patients diagnosed with non-specific pericarditis (coded as 0). Abbreviations: AUC, Area Under the ROC Curve; ROC, Receiver Operating Characteristic.



Figure 3 Decision Curve Analysis of the Nomogram. (A) Decision curve analysis for the nomogram in the training set. (B) Decision curve analysis for the nomogram in the validation set.

patients into low-, medium-, and high-risk groups. Based on decisions made with a high-risk threshold, the misdiagnosis rate was 5.0% (2/40), while the misdiagnosis rate based on physicians' initial clinical decisions was 22.5% (9/40), with a statistically significant difference (P=0.023) (Figure 4A and B). The missed diagnosis rate based on decisions made with a low-risk threshold was 14.3% (6/42). In contrast, the missed diagnosis rate for physicians' initial clinical decisions was 52.4% (22/42), also showing a statistically significant difference (P<0.001) (Figure 4C and D). The rate of overuse of anti-tuberculosis treatment was 77.5% (31/40) (Figure 4E).

## Discussion

In this cohort study, we assessed the value of constructing a diagnostic model for TBP based on indicators such as demographic characteristics, clinical manifestations, and common laboratory results. This diagnostic model can effectively and rapidly identify patients with TBP. Compared to the preliminary clinical decisions made by tuberculosis specialists based on experience, this model significantly reduced the rates of misdiagnosis and missed diagnosis. Based on this model, we recommend that clinicians immediately refer high-risk patients identified for TBP diagnosis to specialized tuberculosis hospitals. Considering the increase in global migration, the spectrum of cardiovascular diseases in developed countries is changing, and physicians in these regions may also have insufficient awareness of tuberculous pericarditis.<sup>10</sup> Therefore, this model can not only play a role in primary healthcare facilities in developing countries but may also provide a powerful tool for physicians in developed countries to diagnose TBP, helping them better meet this challenge.

This study ultimately identified seven variables. Weight loss, history of tuberculosis, history of dust exposure, moderate to severe kidney disease, and ESR are positively correlated with the risk of developing TBP, while BMI and BNP are negatively correlated with the risk of the disease. Nutritional status is considered closely related to the occurrence of tuberculosis. Weight loss reflects the recent rate of nutritional decline, and a low BMI also reflects a state of weakened nutritional status. Malnutrition can impair the human immune system, thereby increasing the risk of tuberculosis infection. A Study have found that individuals with a BMI <18.5 have an estimated risk ratio of 12.4 for developing tuberculosis.<sup>11</sup> At the same time, Mycobacterium tuberculosis infection leads to the activation of TNF- $\alpha$ .<sup>12</sup> TNF- $\alpha$  acts on muscle cells, promoting protein loss through nuclear factor- $\kappa$ B and activating the ubiquitin-proteasome pathway, which further leads to muscle degradation, exacerbating the progression of tuberculosis.<sup>13</sup>

Individuals with a history of tuberculosis are at a significantly higher risk of developing the disease again compared to those without such a history. This increased risk may be due to the reactivation of Mycobacterium tuberculosis present in the body. The bacteria can exist in a dormant state within tissues for months to years, yet still possess the capability to resume growth and cause disease. Immunocompromised states, such as malnutrition, use of corticosteroids, or other



Figure 4 Comparison of Clinical Application Based on the Model and Clinical Experience. (A) Misdiagnosis Due to Model's High-Risk Group Threshold. (B) Misdiagnosis Due to Physicians' Clinical Experience. (C) Missed Diagnoses Due to Model's Low-Risk Group Threshold. (D) Missed Diagnoses Due to Physicians' Clinical Experience. (E) Overuse of Anti-Tuberculosis Treatment.

immunosuppressive agents, can lead to the reactivation of these latent infections.<sup>14,15</sup> Another explanation may be that previous tuberculosis can result in reduced immunity to certain Mycobacterium tuberculosis epitopes, primarily because these epitopes have high homology with bacteria in the microbiome. Anti-tuberculosis treatment may further alter immune responses to these epitopes by affecting the balance of the microbiome.<sup>16</sup> Individuals with a history of dust exposure may exhibit varying degrees of structural changes in lung tissue, such as fibrosis, as well as a reduction in immune defense functions, making them more susceptible to tuberculosis infection.<sup>17,18</sup>

Patients with chronic kidney disease may have impaired functions of various immune cells, including B and T cells, neutrophils, and monocytes. This immune dysfunction not only makes patients more vulnerable to Mycobacterium tuberculosis infection but also may reactivate tuberculosis they may have contracted during childhood or adolescence, turning it into active tuberculosis.<sup>19–22</sup> Consequently, the risk of tuberculosis in patients with chronic kidney disease is much higher than in the general population. The ESR, a method for measuring the sedimentation speed of red blood cells in the blood, is a non-specific test for inflammation and infection. Traditionally used in the diagnosis and assessment of tuberculosis,<sup>23</sup> it reflects the active state of the disease. Patients with active tuberculosis have significantly higher ESR levels than healthy individuals.<sup>24</sup>

BNP, primarily secreted by myocardial cells, is a 32-amino acid residue polypeptide that not only regulates blood pressure and blood volume homeostasis but also has diuretic effects. It is considered the gold standard biomarker for heart dysfunction<sup>25,26</sup> and has a clear diagnostic and prognostic role in heart failure patients.<sup>27</sup> Interestingly, this study found a negative correlation between BNP and the risk of tuberculous pericarditis, which contradicts our usual understanding of cardiopulmonary comorbidities. A possible explanation is that this model compares tuberculous pericarditis with non-specific pericarditis; while tuberculous pericarditis may lead to pericardial effusion, it does not necessarily directly affect cardiac pump function. In contrast, non-specific pericarditis may be more likely to cause cardiac dysfunction, leading to heart failure.<sup>28</sup> Another explanation is that inflammation itself is involved in biological processes including tissue remodeling and metabolism,<sup>29–31</sup> which could affect the secretion of BNP. TBP, as a chronic inflammatory state, may affect the secretion of BNP through different mechanisms, distinct from the acute inflammatory response caused by non-specific pericarditis.

Previous studies<sup>8</sup> have focused on developing diagnostic models for tuberculous pericarditis using indicators such as gamma interferon (IFN- $\gamma$ ), adenosine deaminase (ADA) levels, Mycobacterium tuberculosis polymerase chain reaction detection, and sputum smear and culture. However, many of these indicators are difficult to obtain and time-consuming in primary healthcare settings, making their application scenarios different from the screening model in this study.

The advantage of this study is that the predictive model used predictors that are easily obtainable clinical characteristics and routine laboratory indicators. Compared to traditional microbiological methods, this model not only saved diagnostic time but also may be more cost-effective for high-risk patients in areas with limited access to precision diagnostic tools and economic constraints. However, our study has some limitations. Firstly, this study was a single-center, small-sample study, but it still had good predictive value after internal validation. Additionally, our study lacked external data validation. We plan to conduct prospective studies with multiple centers to further verify the effectiveness of the model.

#### Conclusions

In summary, our model accurately identifies high-risk individuals, reduces the misuse of anti-tuberculosis drugs due to misdiagnosis, and lowers the diagnostic delays and associated adverse outcomes caused by missed diagnoses. We recommend the implementation of this model in primary care to enhance early and accurate diagnosis of TBP, thereby improving patient outcomes and alleviating the burden of this disease in resource-limited or economically underdeveloped regions.

## **Data Sharing Statement**

The raw data supporting the results of this study was uploaded to zenodo (<u>https://zenodo.org/records/13822455</u>) and accessed upon reasonable request.

## **Ethical Approval and Consent to Participate**

All procedures performed in studies involving human participants were following the ethical standards of the institutional and national research committee and with the 1964 helsinki declaration and its later amendments or comparable ethical standards.

## **Consent for Publication**

This study was a retrospective cohort study, and an exemption from the informed consent requirement was approved by the ethics committee of Hangzhou Red Cross Hospital (Ethical Application Ref: 2024YS90).

## **Author Contributions**

All authors made a significant contribution to the work reported, whether that is in the conception, study design, execution, acquisition of data, analysis and interpretation, or in all these areas; took part in drafting, revising or critically reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work. All authors approved the final version of the manuscript.

# Funding

This study was supported by the Zhejiang Administration Bureau of Traditional Chinese Medicine (2024ZR144 and 2023ZR040), the Health Commission of Zhejiang Province (2023KY969), the Hangzhou Municipal Health Commission (A20210115), the 2024 Zhejiang Chinese Medical University Research Project (2024GJYY46), the Hangzhou Biomedical and Health Industry Support Science and Technology Project (2022WJC256), and the Zhejiang Chinese Medicine University Graduate Student Scientific Research Fund Project (2023YKJ15). All sponsors mainly provide remuneration or gratuities for lectures, speeches, manuscript writing, educational activities, or rapid service fee, and do not play any role in study design, data collection, and analysis, or decisions to submit articles for publication.

## Disclosure

Xiaoqun Xu reports article publishing charges and statistical analysis were provided by Zhejiang Administration Bureau of Traditional Chinese Medicine. Xiaoqun Xu reports article publishing charges and statistical analysis were provided by Health Commission of Zhejiang Province. Xiaoqun Xu reports article publishing charges and statistical analysis were provided by Hangzhou Municipal Health Commission. Houyong Zhu reports article publishing charges and statistical analysis were provided by Zhejiang Administration Bureau of Traditional Chinese Medicine. Houyong Zhu reports article publishing charges and statistical analysis were provided by Zhejiang Administration Bureau of Traditional Chinese Medicine. Houyong Zhu reports article publishing charges and statistical analysis were provided by 2024 Zhejiang Chinese Medical University Research Project. Hui Wei reports article publishing charges and statistical analysis were provided by Hangzhou Biomedical and Health Industry Support Science and Technology Project. Chao Yang reports article publishing charges and statistical analysis were provided by Zhejiang Chinese Medicine University Graduate Student Scientific Research Fund Project. The authors report no other conflicts of interest in this work.

## References

- 1. World Health Organization. Global tuberculosis report; 2024. Available from: https://iris.who.int/bitstream/handle/10665/379339/9789240101531eng.pdf?sequence=1. Accessed March 25, 2025.
- 2. Hu X, Xing B, Wang W, et al. Diagnostic values of Xpert MTB/RIF, T-SPOT.TB and adenosine deaminase for HIV-negative tuberculous pericarditis in a high burden setting: a prospective observational study. *Sci Rep.* 2020;10(1):16325. doi:10.1038/s41598-020-73220-y
- 3. Naidoo DP, Laurence G, Sartorius B, et al. The effects of HIV/AIDS on the clinical profile and outcomes post pericardiectomy of patients with constrictive pericarditis: a retrospective review. *Cardiovasc J Afr.* 2019;30(5):251–257. doi:10.5830/CVJA-2019-015
- 4. Mayosi BM, Wiysonge CS, Ntsekhe M, et al. Mortality in patients treated for tuberculous pericarditis in sub-Saharan Africa. *Samj S Afr Med J*. 2008;98(1):36–40.
- 5. Zhou L, Zhang Y, Chai C, et al. Investigation and analysis of tuberculosis health management service projects in Zhejiang Province's basic public health services. *The 34th national academic conference of the Chinese anti-tuberculosis association in 2023 and the forum on the promotion and application of new tuberculosis diagnosis and treatment technologies: a compilation of papers*:2023.
- 6. Cherian G. Diagnosis of tuberculous aetiology in pericardial effusions. Postgrad Med J. 2004;80(943):262-266. doi:10.1136/pgmj.2003.013664
- 7. Cegielski JP, Lwakatare J, Dukes CS, et al. Tuberculous pericarditis in Tanzanian patients with and without HIV infection. *Tuber Lung Dis.* 1994;75 (6):429–434. doi:10.1016/0962-8479(94)90116-3
- 8. Reuter H, Burgess L, van Vuuren W, et al. Diagnosing tuberculous pericarditis. *Qjm-Int J Med.* 2006;99(12):827-839.
- Koppie TM, Serio AM, Vickers AJ, et al. Age-adjusted Charlson comorbidity score is associated with treatment decisions and clinical outcomes for patients undergoing radical cystectomy for bladder cancer. *Cancer-Am Cancer Soc.* 2008;112(11):2384–2392.
- 10. Viljoen CA, Hoevelmann J, Muller E, et al. Neglected cardiovascular diseases and their significance in the Global North. *Herz.* 2021;46 (2):129–137. doi:10.1007/s00059-021-05020-7

- Cegielski JP, Arab L, Cornoni-Huntley J. Nutritional risk factors for tuberculosis among adults in the United States, 1971-1992. Am J Epidemiol. 2012;176(5):409–422. doi:10.1093/aje/kws007
- 12. Chang SW, Pan WS, Lozano BD, et al. Gut hormones, appetite suppression and cachexia in patients with pulmonary TB. *PLoS One*. 2013;8(1): e54564. doi:10.1371/journal.pone.0054564
- Morley JE, Thomas DR, Wilson MM. Cachexia: pathophysiology and clinical relevance. Am J Clin Nutr. 2006;83(4):735–743. doi:10.1093/ajcn/ 83.4.735
- 14. Tufariello JM, Chan J, Flynn JL. Latent tuberculosis: mechanisms of host and bacillus that contribute to persistent infection. *Lancet Infect Dis*. 2003;3(9):578–590. doi:10.1016/S1473-3099(03)00741-2
- 15. Wayne LG, Sohaskey CD. Nonreplicating persistence of mycobacterium tuberculosis. Annu Rev Microbiol. 2001;55:139–163. doi:10.1146/annurev. micro.55.1.139
- Scriba TJ, Carpenter C, Pro SC, et al. Differential recognition of mycobacterium tuberculosis-specific epitopes as a function of tuberculosis disease history. Am J Resp Crit Care. 2017;196(6):772–781. doi:10.1164/rccm.201706-1208OC
- 17. Zhao H, Wang Y, Qiu T, et al. Autophagy, an important therapeutic target for pulmonary fibrosis diseases. *Clin Chim Acta*. 2020;502:139–147. doi:10.1016/j.cca.2019.12.016
- Skowronski M, Halicka A, Barinow-Wojewodzki A. Pulmonary tuberculosis in a male with silicosis. Adv Respir Med. 2018;86(3). doi:10.5603/ ARM.2018.0019
- 19. World Health Organization. Latent Tuberculosis Infection: Updated and Consolidated Guidelines for Programmatic Management. Geneva: World Health Organization; 2018.
- 20. Zuo L, Wang M. Current burden and probable increasing incidence of ESRD in China. Clin Nephrol. 2010;74(Suppl 1):S20-S22.
- Min J, Kwon SK, Jeong HW, et al. End-stage renal disease and risk of active tuberculosis: a nationwide population-based cohort study. J Korean Med Sci. 2018;33(53):e341. doi:10.3346/jkms.2018.33.e341
- 22. Banaga AS, Siddiq NK, Alsayed RT, et al. Prevalence and presentation of tuberculosis among hemodialysis patients in Khartoum, Sudan. Saudi J Kidney Dis T. 2016;27(5):992–996. doi:10.4103/1319-2442.190873
- 23. Krzysztof Bochen AKSM, Dzida G. Erythrocyte sedimentation rate an old marker with new applications. J Pre Clin Res. 2011;2(5):50-55.
- 24. Luo Y, Xue Y, Song H, et al. Machine learning based on routine laboratory indicators promoting the discrimination between active tuberculosis and latent tuberculosis infection. J Infection. 2022;84(5):648–657. doi:10.1016/j.jinf.2021.12.046
- Gardner RS, Ozalp F, Murday AJ, et al. N-terminal pro-brain natriuretic peptide. A new gold standard in predicting mortality in patients with advanced heart failure. *Eur Heart J.* 2003;24(19):1735–1743. doi:10.1016/j.ehj.2003.07.005
- 26. Martineau AR, Newton SM, Wilkinson KA, et al. Neutrophil-mediated innate immune resistance to mycobacteria. J Clin Invest. 2007;117 (7):1988–1994. doi:10.1172/JCI31097
- Goetze JP, Bruneau BG, Ramos HR, et al. Cardiac natriuretic peptides. *Nat Rev Cardiol.* 2020;17(11):698–717. doi:10.1038/s41569-020-0381-0
   Hori K, Kato Y, Suzuki S, et al. Descriptions of etiology, clinical course, and prognosis of patients presenting with pericardial effusion. *Int Heart J.* 2024;65(3):452–457. doi:10.1536/ibi.23-511
- Ong SB, Hernández-Reséndiz S, Crespo-Avilan GE, et al. Inflammation following acute myocardial infarction: multiple players, dynamic roles, and novel therapeutic opportunities. *Pharmacol Therapeut*. 2018;186:73–87. doi:10.1016/j.pharmthera.2018.01.001
- 30. Ewing M, Karper JC, de Vries MR, et al. Editorial: inflammation and immunomodulation in cardiovascular remodeling. *Front Cardiovasc Med.* 2022;9:1094094. doi:10.3389/fcvm.2022.1094094
- Daou D, Gillette TG, Hill JA. Inflammatory mechanisms in heart failure with preserved ejection fraction. *Physiology*. 2023;38(5). doi:10.1152/ physiol.00004.2023

Journal of Inflammation Research



#### Publish your work in this journal

The Journal of Inflammation Research is an international, peer-reviewed open-access journal that welcomes laboratory and clinical findings on the molecular basis, cell biology and pharmacology of inflammation including original research, reviews, symposium reports, hypothesis formation and commentaries on: acute/chronic inflammation; mediators of inflammation; cellular processes; molecular mechanisms; pharmacology and novel anti-inflammatory drugs; clinical conditions involving inflammation. The manuscript management system is completely online and includes a very quick and fair peer-review system. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/journal-of-inflammation-research-journal

🖪 🗙 in 🗖

4693